% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P1.38}{24(1)}{2017}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}



%%%%%%%
\usepackage[usenames,dvipsnames]{color}
\usepackage{bm}
\usepackage[mathscr]{euscript}
\def\dessin#1#2{\includegraphics[#1]{#2}}
%%%%%%%%

% use this command to create hyperlinks (optional and recommended)

\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}


% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

%%%%%%%%%%%%%%%%%%%%%% My Macros




\def\auteur#1{{\sc #1}}
\def\titreref#1{{\em #1}}
\def\vol#1{{\bf #1}}
\def\pref#1{{\rm (\ref{#1})}}
\def\define#1{{\bf{#1}}}
\def\petit#1{\hbox{\scriptsize$#1$}}


\newcommand{\SymGroup}{\mathbb{S}}
\newcommand{\fix}{\operatorname{fix}}
\def\qfoulkes#1#2#3#4{\langle #1,#2,#3,#4\rangle_q}
\def\foulkes#1#2#3#4{\langle #1,#2,#3,#4\rangle}
\def\GL{\mathrm{GL}}
\newcommand{\qbinom}[2]{\genfrac{(}{)}{0pt}{}{#1}{#2}_q}
\def\mod{\,\ \mathbf{mod}\,}
\def\sign{\,\ \mathrm{sign}}
\DeclareMathOperator{\tenseur}{\otimes}
\def\P#1{(h_2+e_2)^{#1}}
\def\Q#1{(h_2-e_2)^{#1}}
\def\N{\mathbb{N}}
\def\C{\mathbb{C}}
\def\Z{\mathbb{Z}}
\def\Rat{\mathbb{Q}}
\newcommand{\scalar}[2]{{\langle#1,#2 \rangle}}
\def\charac{\raise 2pt\hbox{\large$\chi$}}
\def\Id{\mathrm{Id}}


% if needed include a line break (\\) at an appropriate place in the title

\title{\bf A $q$-Analog of Foulkes' Conjecture}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address 

\author{Fran\c{c}ois Bergeron\thanks{Supported by NSERC.}\\
\small D\'epartement de Math\'ematiques\\[-0.8ex]
\small Universit\'e du Qu\'ebec \`a Montr\'eal\\[-0.8ex] 
\small Canada\\
\small\tt bergeron.francois@uqam.ca}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Mar 11, 2016}{Feb 8, 2017}{Feb 17, 2017}\\
\small Mathematics Subject Classifications: 05E05, 05E10, 20C30, 20C33}


\begin{document}
\begin{thebibliography}{99}  

\bibitem{bergeron}
F.~Bergeron.
 \newblock
 Algebraic Combinatorics and Coinvariant Spaces.  
 \newblock {\em CMS Treatise in Mathematics}, CMS and A.K.Peters,  2009.


\bibitem{species}
F.~Bergeron, P.~Leroux, and G.~Labelle.
 \newblock
 Combinatorial Species and Tree-Like structures.
 \newblock
 {\em Encyclopedia of Mathematics and its Applications}, 
 Volume 67, Cambridge University Press, 1998.


\bibitem{briand}
E.~Briand.
 \newblock
Polyn\^omes multisym\'etriques, PhD dissertation.
 \newblock
 University Rennes I, Rennes, France, 2002.

\bibitem{brion}
M.~Brion.
 \newblock
Stable properties of plethysm: on two conjectures of Foulkes.
 \newblock
{\em Manuscripta Mathematica},  80:347--371,1993.

\bibitem{deBoeck}
M.~de Boeck.
 \newblock
Relationships between plethysm coefficients.
 \newblock	
See \href{http://arxiv.org/abs/1409.0734}{arXiv:1409.0734}.


\bibitem{cheung}
M.-W.~Cheung, C.~Ikenmeyer, and S.~Mkrtchyan.
 \newblock
Symmetrizing Tableaux and the 5th case of the Foulkes Conjecture.
 \newblock
{\em Journal of Symbolic Computation},  80(3): 833--843, 2017.
\newblock
    See \href{http://arxiv.org/abs/1509.03944}{arXiv:1509.03944}. 

\bibitem{colmenarejo}
L.~Colmenarejo.
 \newblock
Stability Properties of the Plethysm: a Combinatorial Approach.
 \newblock
{\em DMTCS Proceedings FPSAC'15} 877--888, 2015.
 \newblock
See \href{http://arxiv.org/abs/1505.03842}{arXiv:1505.03842}. 

\bibitem{doran}
W.~F.~Doran IV.
 \newblock
On Foulkes' conjecture.
 \newblock
{\em Journal of Pure and Applied Algebra},  130:85--98,1998.

\bibitem{hadamard}
J.~Hadamard.
\newblock
Sur les conditions de d\'ecomposition des formes.
\newblock
{\em Bull. Soc. Math. France} 27: 34--47, 1899.


\bibitem{foulkes}
H.~O.~Foulkes.
 \newblock
Concomitants of the quintic and sextic up to degree four in the coefficients of the ground form.
 \newblock
{\em Journal of the London Mathematical Society},  25:205--209,1950.

\bibitem{hermite}
C.~Hermite.
 \newblock
Sur la th\'eorie des fonctions homog\`enes \`a deux ind\'etermin\'ees.
 \newblock
{\em Cambridge and Dublin Mathematical Journal},  9:172--217,1894.

\bibitem{Ikenmeyer}
C.~Ikenmeyer.
 \newblock
On McKay's propagation theorem for the Foulkes conjecture.
\newblock
    See \href{https://arxiv.org/abs/1509.04957}{ arXiv:1509.04957v1}. 


\bibitem{Kahle}
T.~Kahle and M.~Michalek.
 \newblock
Plethysm and lattice point counting.
 \newblock
{\em Foundations of Computational Mathematics} 16(5):1241--1261, 2016.
\newblock
    See \href{https://arxiv.org/abs/arXiv:1408.5708}{arXiv:1408.5708}
    
\bibitem{Landsberg}
J.M. Landsberg.
\newblock
Geometric Complexity Theory: an introduction for geometers.
\newblock
{\em J. Ann Univ Ferrara} 61(1):65--117.
\newblock
   See \href{https://arxiv.org/abs/1305.7387}{arXiv:1305.7387}

    
\bibitem{littlewood}
D.~E.~Littlewood.
 \newblock
Invariant theory, tensors and group characters.
 \newblock
{\em Philosophical Transactions of the Royal Society of London}, (A) 305--365,1944.
    
\bibitem{macdonald} 
I.~G. Macdonald.
 \newblock
Symmetric functions and Hall polynomials, Second Edition.
 \newblock
{\em Clarendon Press}, Oxford, 1995.
  
\bibitem{manivelsylvester}
L.~Manivel.
 \newblock
An extension of the Cayley-Sylvester formula.
 \newblock
{\em European Journal of Combinatorics}, 28:1839--1842, 2007.

\bibitem{manivelgaussian}
L.~Manivel.
 \newblock
Gaussian maps and plethysm.
 \newblock
In {\em Algebraic geometry} (Catania, 1993/Barcelona, 1994), Volume 200 of {\em Lecture Notes in Pure and Applied Mathematics}, 91--117. Dekker, New York, 1998.


\bibitem{mckay}
T.~McKay.
 \newblock
On plethysm conjectures of Stanley and Foulkes.
 \newblock
{\em Journal of Algebra},  319(5):2050--2071, 2008.

 \bibitem{wildon}
 R.~Paget and M.~Wildon.
 \newblock
Minimal and maximal constituents of twisted Foulkes characters.
 \newblock
{\em Journal of the London Mathematical Society}.  93(2):301--318, 2016.
 
 \bibitem{paget}
R.~Paget and M.~Wildon.
 \newblock
Set Families and Foulkes Modules.
 \newblock
{\em Journal of Algebraic Combinatorics}, 34(3):525--544, 2011.
See \href{http://arxiv.org/abs/1007.2946v2}{arXiv:1007.2946v2}.
 
 \bibitem{pylyavskyy}
P.~Pylyavskyy.
 \newblock
On Plethysm Conjectures of Stanley and Foulkes: the $2\times n$ Case.
 \newblock
{\em The Electronic Journal of Combinatorics} 11(2) (The Stanley Festschrift volume):\href{http://www.combinatorics.org/ojs/index.php/eljc/article/view/v11i2r8}{\#R8},   2004-2006.

 
 
 \bibitem{sivek}
S.~Sivek.
 \newblock
Some plethysm results related to Foulkes' conjecture.
 \newblock
{\em The Electronic Journal of Combinatorics} 13:\href{http://www.combinatorics.org/ojs/index.php/eljc/article/view/v13i1r24}{\#R24},  2006. 
 
 \bibitem{vessenes}
R.~Vessenes.
 \newblock
Generalized Foulkes' Conjecture and tableaux.
 \newblock
{\em Journal of Algebra}, 277:579--614, 2004.
 
 \bibitem{weintraub}
S.~H.~Weintraub.
 \newblock
Some observations on plethysms.
 \newblock
{\em Journal of Algebra}, 129:103--114, 1990.

\end{thebibliography}


\maketitle
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{abstract} We propose a $q$-analog of classical plethystic  conjectures due to Foulkes. In our conjectures, a divided difference of plethysms of Hall-Littlewood polynomials $H_n(\bm{x};q)$ replaces the analogous difference of plethysms of complete homogeneous symmetric functions $h_n(\bm{x})$ in Foulkes' conjecture. At $q=0$, we get back the original statement of Foulkes, and we show that our version holds at $q=1$.  We discuss further supporting evidence, as well as various generalizations, including a $(q,t)$-version.

 \bigskip\noindent \textbf{Keywords:} Foulkes' conjecture; Macdonald polynomials; $q$-analog
\end{abstract}


\maketitle  \tableofcontents



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction} 
The Foulkes' conjecture, which dates back to 1950 (see~\cite{foulkes}), has a long and interesting history. Some headway has been made on it, but it remains open in general. A survey of the current state of affairs can be found in~\cite{cheung,Landsberg}, and related papers include \cite{briand, deBoeck, colmenarejo, manivelsylvester, manivelgaussian, 
mckay, weintraub}. In its original form, the conjecture  states that, for any positive integers $a$ and $b$, with $a<b$, the difference of plethysms\footnote{This is an important operation on symmetric functions which was introduced by Littlewood~\cite{littlewood}. Its explicit definition is recalled further below.}  of complete homogeneous symmetric polynomials
\begin{equation}\label{foulkes_conj}
    h_b\circ h_a- h_a\circ h_b,
 \end{equation}
expands with positive (integer) coefficients in the Schur basis $\{s_\mu\}_{\mu\vdash n}$ of symmetric polynomials (here $\mu$ runs over the set of partitions of $n=a\,b$). Instances of this positivity are 
  $$ h_4\circ h_2- h_2\circ h_4=s_{{422}}+s_{{2222}},\qquad {\rm and}\qquad
     h_4\circ h_3- h_3\circ h_4=s_{{732}}+s_{{5421}}+s_{{6222}}.$$
From the point of view of representation theory, one may interpret Foulkes' conjecture as saying that there is a $\GL(V)$-module inclusion of  the composite of symmetric powers $S^a(S^b(V))$ inside $S^b(S^a(V))$. Therefore each $\GL(V)$-irreducible occurs with smaller multiplicity in $S^a(S^b(V))$ than it does in $S^b(S^a(V))$, and the conjecture reflects this at the level of the corresponding characters (with Schur polynomials appearing as characters of irreducible representations).
Many interesting point of view may be considered, and some of these are nicely discussed in~\cite{Landsberg} both with an historical perspective\footnote{Where it is underlined that the question raised by Foulkes apparently goes back to Hadamard~\cite{hadamard}.}, and explanations of ties with Geometric Complexity Theory.

Here, we consider symmetric ``polynomials'' (we will often say function) in a denumerable set of variables $\bm{x}=x_1,x_2,x_3,\ldots$, which are typically not mentioned explicitly.
This makes it so that our statements hold irrespective of the number of variables occurring in the symmetric functions considered ({\sl ergo} the dimension of the vector space $V$). 
Notice that the case of Foulke's conjecture when this dimension is $2$ is a theorem. Indeed, it corresponds to Hermite's {\em Law of reciprocity} (see~\cite{hermite}), which says that the modules $S^a(S^b(V))$ and $S^b(S^a(V))$ coincide when $V$ has dimension $2$ (over $\C$). Another fact that is worth recalling is that Brion~\cite{brion} has shown that \pref{foulkes_conj} holds if $b$ is large enough with respect to $a$. 


Our $q$-analog replaces the relevant homogeneous symmetric function $h_n=h_n(\bm{x})$  by the Macdonald  (Hall-Littlewood) polynomial
    $$H_n(\bm{x};q):=\sum_{\mu\vdash n} K_{\mu}(q) s_\mu(\bm{x}),
    \qquad{\rm where}  \qquad K_{\lambda}(q)=\sum_\tau q^{c(\tau)},$$
  with $\tau$ running through the set of standard tableaux of shape $\lambda$, and $c(\tau)$ standing for the charge statistic. As we will recall further below, one has $H_n(\bm{x};0)=h_n(\bm{x})$. Hence we are considering a slightly different notion of $q$-analog, in which it is the specialization at $q=0$ (rather than $q=1$) that gives back the original statement.
\begin{conjecture}[$q$-Foulkes]\label{conjqFoulkes}
For any integers $0<a\leq b$, the Schur function expansion of the divided difference
\begin{equation}\label{qFoulkesconj}
  F_{a,b}(\bm{x};q):= \frac{H_b\circ H_a  -H_a\circ H_b}{1-q}
\end{equation}
 has coefficients in $\N[q]$.
\end{conjecture}
\noindent For short, one says that  $F_{a,b}(\bm{x};q)$ is {\em Schur positive}. This has been checked to hold whenever $ab\leq 25$, and we will prove in the sequel  that the conjecture is true at $q=1$. We have the specialization $F_{a,b}(\bm{x};0)=f_{a,b}:=h_b\circ h_a-h_a\circ h_b$.
We will show later that it makes sense to divide by $1-q$. 
For instance, with $a=2$ and $b=3$, we find after calculation that  expression \pref{qFoulkesconj} expands in the Schur basis as as
\begin{eqnarray*}  F_{2,3}(\bm{x};q)&=&
\left( {q}^{2}+2\,{q}^{3}+{q}^{4} \right) s_{{33}}+ \left( 1+q+{q}^{2
}+{q}^{3} \right) s_{{222}}\\
&&\ + \left( q+2\,{q}^{2}+2\,{q}^{3}+2\,{q}^{4}
+{q}^{5} \right) s_{{321}}\\
&&\ + \left( q+2\,{q}^{2}+3\,{q}^{3}+3\,{q}^{4}+
2\,{q}^{5}+{q}^{6} \right) s_{{2211}}\\
&&\ + \left( {q}^{2}+2\,{q}^{3}+{q}^{
4} \right) s_{{3111}}+ \left( {q}^{2}+2\,{q}^{3}+3\,{q}^{4}+2\,{q}^{5}
 \right) s_{{21111}}\\
&&\ + \left( {q}^{3}+{q}^{4}+{q}^{5}+{q}^{6} \right) s
_{{111111}}.
\end{eqnarray*} 
This does specialize, at $q=0$, to the corresponding case of Foulkes' conjecture:
\begin{displaymath}
   f_{2,3}=h_3\circ h_2-h_2\circ h_3=s_{222}
 \end{displaymath} 
 
  A second part of Foulkes' conjecture,  shown to be true by Brion~\cite{brion}, concerns the stability of coefficients as $b$ grows while $a$ remains fixed. To simplify its statement, we consider the linear operator which sends a Schur function $s_\mu(\bm{x})$ to $s_{\overline{\mu}}(\bm{x})$, where $\overline{\mu}$ is the partition obtained by removing the largest part in $\mu$. Let us write $\overline{f}$ for the effect of this operator on a symmetric function $f$. 
For example, we get
   $\overline{s_{622}+s_{442}+s_{4222}+s_{22222}} =s_{22}+s_{42}+s_{222}+s_{2222}$,
  which is clearly not homogeneous.
Using this notation convention, the second part of Foulkes' conjecture states that, for all $a\leq b$, the Schur expansion of
\begin{equation}\label{foulkes2}
  \overline{f_{a,b+1}}-\overline{f_{a,b}}= \overline{(h_{b+1}\circ h_a-h_a\circ h_{b+1})} -\overline{ (h_{b}\circ h_a-h_a\circ h_{b})}
\end{equation}
also affords positive integers polynomials as coefficients. 
Observe that the ``Bar'' operator allows the comparison of  homogeneous functions of different degrees, namely $f_{a,b+1}$ of degree $a(b+1)$ with $f_{a,b}$ of degree $ab$.
For instances of, one calculates $ \overline{f_{2,4}}- \overline{f_{2,3}}=s_{222}$, since the left-hand side is equal to $\overline{(s_{422}+s_{2222})} -\overline{s_{222}}$, in which $\overline{s_{422}}$ cancels out with $\overline{s_{222}}$.

In~\cite{brion}, Brion has shown \pref{foulkes2} reducing (see Appendix for notations) it to showing that 
   $$\scalar{h_a\circ h_b}{s_\lambda}\leq \scalar{h_a\circ h_{b+1}}{s_{\lambda+(a)}},$$
where the \define{sum} $\lambda+\mu$ of two partitions $\lambda$ and $\mu$, is the partition whose parts are $\lambda_i+\mu_i$ (with the convention that $\lambda_i=0$ if $i$ greater than the number of parts of $\lambda$). More results along these lines may be found in \cite{Ikenmeyer, mckay}.
A similar phenomenon also seems to hold in our context, leading us to state the following.
\begin{conjecture}[$q$-stability]
For any integers $0<a\leq b$, and any Schur function $s_\lambda$, we have
\begin{equation}\label{qFoulkesconjstab}
     \langle\overline{F_{a,b+1}(\bm{x};q)}-\overline{F_{a,b}(\bm{x};q)},s_\lambda(\bm{x})\rangle\in \N[q].
  \end{equation}
\end{conjecture}
\noindent
The smallest non-trivial example is:
\begin{eqnarray*}
\overline{F_{2,4}(\bm{x};q)}-\overline{F_{2,3}(\bm{x};q)}&=&(1+q)\Big(\left( \petit{{q}^{3}+2\,{q}^{4}+2\,{q}^{5}+{q}^{6}} \right) s_{{3}}+ \left(\petit{{q}^{2}+2\,{q}^{3}+3\,{q}^{4}+3\,{q}^{5}+2\,{q}^{6}+{q}^{7}} \right) s_
{{21}}\\
&& + \left(\petit{ {q}^{3}+2\,{q}^{4}+2\,{q}^{5}+{q}^{6}} \right) s_{{111}}
+ \left(\petit{ {q}^{2}+2\,{q}^{4}+{q}^{6} }\right) s_{{4}}\\
&& + \left( \petit{q+2\,{q}^{2}+6\,{q}^{3}+7\,{q}^{4}+9\,{q}^{5}+6\,{q}^{6}+4\,{q}^{7}+{q}^{8}} \right) s_{{31}}\\
&& + \left( \petit{q+3\,{q}^{2}+4\,{q}^{3}+7\,{q}^{4}+5\,{q}^{5
}+6\,{q}^{6}+2\,{q}^{7}+2\,{q}^{8} }\right) s_{{22}}\\
&& + \left( \petit{2\,{q}^{2}
+6\,{q}^{3}+10\,{q}^{4}+13\,{q}^{5}+11\,{q}^{6}+8\,{q}^{7}+3\,{q}^{8}+
{q}^{9}} \right) s_{{211}}\\
&& + \left( \petit{{q}^{3}+4\,{q}^{4}+6\,{q}^{5}+7\,{q}
^{6}+4\,{q}^{7}+2\,{q}^{8} }\right) s_{{1111}}\\
&& + \left( \petit{q+2\,{q}^{2}+5\,
{q}^{3}+6\,{q}^{4}+8\,{q}^{5}+6\,{q}^{6}+5\,{q}^{7}+2\,{q}^{8}+{q}^{9}}
 \right) s_{{32}}\\
&& + \left( \petit{3\,{q}^{2}+4\,{q}^{3}+10\,{q}^{4}+9\,{q}^{5}
+11\,{q}^{6}+6\,{q}^{7}+4\,{q}^{8}+{q}^{9} }\right) s_{{311}}\\
&& + \left( \petit{2
\,q+4\,{q}^{2}+9\,{q}^{3}+12\,{q}^{4}+15\,{q}^{5}+13\,{q}^{6}+11\,{q}^
{7}+6\,{q}^{8}+3\,{q}^{9}+{q}^{10}} \right) s_{{221}}\\
&& + \left( \petit{2\,{q}^{2
}+6\,{q}^{3}+11\,{q}^{4}+16\,{q}^{5}+17\,{q}^{6}+14\,{q}^{7}+9\,{q}^{8
}+4\,{q}^{9}+{q}^{10}} \right) s_{{2111}}\\
&& + \left( \petit{{q}^{3}+3\,{q}^{4}+6
\,{q}^{5}+7\,{q}^{6}+8\,{q}^{7}+5\,{q}^{8}+3\,{q}^{9}+{q}^{10}}
 \right) s_{{11111}}\\
&& +{  \left( \petit{1+2\,{q}^{2}+{q}^{3}+4\,{q}^{4}+2\,{q}^{5
}+5\,{q}^{6}+{q}^{7}+3\,{q}^{8}+{q}^{10}} \right) s_{{222}}}\\
&& +\left( \petit{q+{
q}^{2}+4\,{q}^{3}+5\,{q}^{4}+9\,{q}^{5}+8\,{q}^{6}+9\,{q}^{7}+5\,{q}^{
8}+4\,{q}^{9}+{q}^{10}+{q}^{11}} \right) s_{{2211}}\\
&& + \left( \petit{{q}^{2}+{q}
^{3}+5\,{q}^{4}+5\,{q}^{5}+9\,{q}^{6}+7\,{q}^{7}+7\,{q}^{8}+3\,{q}^{9}
+2\,{q}^{10}}\right) s_{{21111}}\\
&& + \left( \petit{{q}^{3}+{q}^{4}+3\,{q}^{5}+3
\,{q}^{6}+4\,{q}^{7}+3\,{q}^{8}+3\,{q}^{9}+{q}^{10}+{q}^{11}} \right) s
_{{111111}}\\
&& + \left( \petit{{q}^{4}+{q}^{6}+{q}^{8}+{q}^{10}} \right) s_{{
1111111}}\Big),
\end{eqnarray*}
and we do observe that this specializes to the (much simpler) classical Foulkes case when we set $q=0$. Another stability, in the vein of Manivel~\cite{manivelgaussian},  that seems to hold in our $q$-context is that
   $$(\overline{F_{a+1,b+1}(\bm{x};q)}-\overline{F_{a+1,b}(\bm{x};q)})-
      (\overline{F_{a,b+1}(\bm{x};q)}-\overline{F_{a,b}(\bm{x};q)}),$$
is Schur positive, for all $a<b$.

\subsection*{Extensions}\label{extensions}
In her thesis, partly presented in~\cite{vessenes},  Vessenes attributes\footnote{However, it is not clear in~\cite{doran} where this exact statement can be found, if at all.} the following generalization of Foulkes' conjecture to Doran~\cite{doran}. As before, let $a<b$ and consider two extra integers $c$ and $d$, both lying between $a$ and $b$, such that $ab=cd$ (observe that we do not assume that $c<d$).
 Then Vessenes's extension of Foulkes' conjecture states that
    \begin{equation}\label{doran}
         h_c\circ h_d-h_a\circ h_b
    \end{equation}
 expands with positive coefficients in the Schur basis.
Setting $c=b$ and $d=a$ clearly gives back the original statement of Foulkes. 
For example, one calculates that
    $$h_3\circ h_4-h_2\circ h_6=s_{{93}}+s_{{444}}+s_{{642}}+s_{{741}}+s_{{822}}.$$
Vessenes proves that \pref{doran} is indeed Schur positive whenever $a=2$, and more instances may be shown to hold using formulas of \cite{Kahle} for the plethysms $s_\mu\circ h_k$, for partitions $\mu$ of $3$, $4$ and $5$. We have also checked it explicitlly for all cases when the overall degree is less or equal to $36$.

Extensive computer algebra experiments\footnote{For all cases when the overall degree if less or equal to $32$.} suggest that there are two natural (new) extensions to this conjecture, namely
\begin{conjecture}\label{conjkDoran} Assume that $a\leq c,d\leq b$, with $ab=cd$, and $k$ is any positive integer, then we have Schur positivity of both the differences
\begin{equation}\label{kdoran}
  {h_c\circ (h_d^k)-h_a\circ (h_b^k)},\qquad {\rm and}\qquad  h_c\circ s_\delta-h_a\circ s_\beta,
  \end{equation}
where we consider the rectangular shape partitions 
   $$\delta=\underbrace{dd\cdots d}_k,\qquad {\rm and}\qquad \beta=\underbrace{bb\cdots b}_k.$$
\end{conjecture}
\noindent
More experiments (for all cases when the overall degree is less than $18$) suggest that all the above conjectures are all encompassed in the following one, which involves the (combinatorial) two parameters Macdonald polynomials $H_\mu(\bm{x};q,t)$ indexed by rectangular shape partitions.
\begin{conjecture}\label{conjqt} Assume that $a\leq c,d\leq b$, with $ab=cd$, and $k$ is any positive integer. Then, we have $\N[q]$-Schur positivity of
\begin{equation}\label{conjqtDoranformule}
   \frac{H_c\circ H_\delta(\bm{x};q,t)-H_a\circ H_\beta(\bm{x};q,t)}{1-q},\qquad {\rm with}\qquad
    \delta=\underbrace{dd\cdots d}_k,\quad {\rm and}\quad \beta=\underbrace{bb\cdots b}_k.  \end{equation}
\end{conjecture}
\noindent
We will show why division by $1-q$ makes sense and how it implies previous conjectures in the following section. It is interesting to observe that, at $t=1$, formula~\pref{conjqtDoranformule} specializes to the nice $q$-analog
\begin{equation}\label{conjqtk}
   F_{a,b;c,d}^{(k)}(\bm{x};q):= \frac{H_c\circ H_d^k(\bm{x};q)-H_a\circ H_b^k(\bm{x};q)}{1-q},\end{equation}
 which thus have to be Schur positive if~Conjecture~\ref{conjqt} is to hold.
 


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Supporting facts and implications}\label{resultats}
To further discuss and exploit the several implications of conjectures \ref{conjqFoulkes} and \ref{conjqt}, and their ties to 
conjecture \ref{conjkDoran}, 
we rapidly recall some properties and specializations of the symmetric polynomials $H_\mu(\bm{x};q,t)$. 
\subsection*{Combinatorial Macdonald polynomials} Recall that the polynomial $H_n(\bm{x};q)$ is a special instance of the ``combinatorial'' Macdonald polynomials $H_\mu=H_\mu(\bm{x};q,t)$. Since the parameter $t$ only appears in the $H_\mu$'s that are indexed by partitions having at least $2$ parts, it makes sense to avoid its mention in $H_n$. For instance, we have
\begin{eqnarray*}
   H_3(\bm{x};q,t)&=&s_{{3}}(\bm{x})+ \left( {q}^{2}+q \right) s_{{21}}(\bm{x})+{q}^{3}s_{{111}}(\bm{x}),\\
H_{21}(\bm{x};q,t)&=&s_{{3}}(\bm{x})+ \left( q+t \right) s_{{21}}(\bm{x})+q\,t\,s_{{111}}(\bm{x}),\\
H_{111}(\bm{x};q,t)&=&s_{{3}}(\bm{x})+ \left( {t}^{2}+t \right) s_{{21}}(\bm{x})+{t}^{3}s_{{111}}(\bm{x}).
\end{eqnarray*}
The $H_\mu$ are orthogonal with respect to the scalar product defined on the power-sum basis by the formula
    $$\langle p_\mu(\bm{x}),p_\mu(\bm{x})\rangle_{q,t}:=(-1)^{n-\ell(\mu)}\,z_\mu\,\prod_{k\in \mu} (1-q^k)(1-t^k),$$
 with $\langle p_\mu(\bm{x}),p_\lambda(\bm{x})\rangle_{q,t}=0$ whenever $\mu\not=\lambda$. Here, $\ell(\mu)$ stands for the number of parts of $\mu$, and $k$ runs over the parts of $\mu$. They afford the following specializations:
\begin{eqnarray}
   {\rm a)}&&\ H_\mu(\bm{x};0,0) = h_n(\bm{x}),\label{H00}\\
   {\rm b)}&&\ H_\mu(\bm{x};1,1) = h_1^n(\bm{x}),\label{H11}\\
   {\rm c)}&&\ H_\mu(\bm{x};0,1) = h_\mu(\bm{x}),\label{H01}\\
   {\rm d)}&&\ H_\mu(\bm{x};0,t)\big|_{t^{\rm maxdeg}} = s_\mu(\bm{x}).\label{H0max}
\end{eqnarray}
Furthermore we have
    \begin{equation}\label{Hmuq1}
        H_\mu(\bm{x};q,1)=\prod_{k\in \mu}H_k(\bm{x};q),\qquad{\rm with}\qquad H_k(\bm{x};q)=\prod_{i=1}^n(1-q^i)\, h_k\left[\frac{\bm{x}}{1-q}\right],
    \end{equation}
 which may be written in the following form
using plethystic rules of calculation, and formula~\pref{hpexpansion}:
   \begin{equation}\label{defhnpower}
       H_n(\bm{x};q) =\sum_{\mu\vdash n}\frac{[n]_q!\,}
           {z_\mu\,[\mu_1]_q[\mu_2]_q\cdots [\mu_\ell]_q}\, (1-q)^{n-\ell(\mu)}\ p_\mu(\bm{x}),
   \end{equation}
with $\ell=\ell(\mu)$ standing for the number of parts of $\mu$.  
We also have the symmetry
      \begin{equation}\label{echange}
        H_{\mu'}(\bm{x};q,t)=H_\mu(\bm{x};q,t).
       \end{equation}
Hence it follows that
   \begin{equation}\label{Hmuq1t}
        H_\mu(\bm{x};1,t)=\prod_{k\in \mu'}H_k(\bm{x};t).
    \end{equation}
Now, we may show that division by $1-q$ makes sense in formulas \pref{qFoulkesconj} and \pref{conjqtDoranformule}.  Indeed, for any $a$ and $b$, formula~\pref{H11} implies that we have $(H_a\circ H_b)(\bm{x};1)= h_1(\bm{x})^{ab}$, since the evaluation at $q=1$ is compatible\footnote{Meaning that they commute as operators. Observe that this is not so with the evaluation at $q=-1$.} with plethysm. Hence, the  numerator of the right-hand side of \pref{qFoulkesconj} vanishes at $q=1$, and is thus divisible by $1-q$.   
More generally, using \pref{Hmuq1} with $\mu=bb\cdots b$ and hence  $$\mu'=\underbrace{kk\cdots k}_b,$$
one sees that
 $$ H_a\circ H_{bb\cdots b} (\bm{x};1,t)=
         h_1^a\circ(H_k(\bm{x};t)^b)=H_k(\bm{x};t)^{ab},$$
 it follows that the coefficients of each Schur function in the expression  of~\pref{conjqtDoranformule} are polynomials in $q$.
 
Several specializations of \pref{conjqtDoranformule} are of interest. In particular, at $q=0$, we get
   $$h_a\circ H_{bb\cdots b}(\bm{x};0,t)\preceq_s h_c\circ H_{dd\cdots d}(\bm{x};0,t),	$$
 where we write $f\preceq_s g$ if $g-f$ is Schur positive.
Comparing coefficients of the highest power of $t$ on both sides of this gives 
   $$h_a\circ s_{bb\cdots b} \preceq_s h_c\circ s_{dd\cdots d},$$
 which is precisely the second statement in~\pref{kdoran}.
Likewise, setting $t=1$ we get
   $$h_a\circ (h_b^k) \preceq_s h_c\circ (h_d^k),$$
Specializing at $q=0$ and $t=0$ also gives the following instance of \pref{doran}
   $$h_a\circ h_{(kb)} \preceq_s h_c\circ h_{(kd)}.$$
We will also consider in the sequel the following\footnote{This is well-known in the theory of Macdonald polynomials, and all properties also mentioned.} $q$-analog of Schur functions:
    $$S_\mu(\bm{x};q):=\omega\, q^{n(\mu')}\,H_\mu(\bm{x};1/q,0),$$
defined in terms of specialization at $t=0$ of the combinatorial Macdonald polynomials\footnote{See Appendix A for various notations used here.} $H_\mu(\bm{x};q,t)$, with $\omega$ standing for the ``usual'' linear involution that sends $s_\mu(\bm{x})$ to $s_{\mu'}(\bm{x})$.
 In particular, one may check that $S_\mu(\bm{x};0)=s_\mu(\bm{x})$ and $S_\mu(\bm{x};1)=e_{\mu'}(\bm{x})$. 
For instance, we have
      $$S_{32}(\bm{x};q)=s_{{32}}(\bm{x})+q\,s_{{311}}+q \left( q+1 \right) s_{{221}}(\bm{x})+{q}^{2} \left( q+1
 \right) s_{{2111}}(\bm{x})+{q}^{4}s_{{11111}}(\bm{x}),$$
 and $S_{32}(\bm{x};1)=e_1(\bm{x})\,e_2(\bm{x})^2$.
 Observe that all terms in the Schur expansion of $S_\mu(\bm{x};q):$ are indexed by partitions that are dominated by $\mu$.
Moreover we get back our previous context for $\mu=(a)$, since $S_{(a)}(\bm{x};q)=H_a(\bm{x},q)$.

   
  
  %%%%%%%%%%%%%%%%%%%
\subsection*{Dimension Count}
As mentioned previously, when a homogeneous degree $n$ symmetric function $f$ occurs as a (graded) Frobenius transform of the character of a $\SymGroup_n$-module, the dimension (Hilbert series) of this module may be readily calculated
by taking its scalar product with $h_1^n$. On the other hand, general principles insure that there exists such a module (albeit not explicitly known) whenever $f$ expands positively (with coefficients in $\N[q]$) in the Schur function basis. Finding an explicit formula for this ``dimension'' may give a clue on what kind of module one should look for in order to prove the conjectures.
With this in mind, let us set the notation 
   $$\dim(f):=\langle h_1^n,f\rangle.$$
For instance, we may easily calculate that
\begin{equation}\label{partionab}
    \dim(h_a\circ h_b) = \frac{(ab)!}{a!\, b!^a},
 \end{equation}
since $h_1^{ab}$ may only occur in the plethysm $h_a\circ h_b$ as
     $$\frac{h_1^a}{a!}\left[\frac{h_1^b}{b!}\right]= \frac{h_1^{ab}}{a!\, b!^a}. $$
In a classical combinatorial setup, formula~\pref{partionab} is easily interpreted as the number of partitions of a set of cardinality $ab$, into blocks each having size $b$. We say that this is a $b^a$-partition.
Indeed, using a general framework such as the Theory of Species (see~\cite{species}), it is well understood that $h_a\circ h_b$ may be  interpreted as the Polya cycle index enumerator of such partitions, i.e.:
    $$h_a\circ h_b  = \frac{1}{n!}\sum_{\sigma\in \SymGroup_{n}}
            \fix_\sigma p_1^{d_1}p_2^{d_2}\cdots p_n^{d_n},$$
   where $n=ab$, and $d_k$ denotes the number of cycles of size $k$ in $\sigma$. Here, we further denote by $ \fix_\sigma$ the number of $b^a$-partitions that are fixed by a permutation $\sigma$, of the underlying elements. It follows that 
\begin{equation}\label{dimFoulkes}
    \dim(F_{a,b}(\bm{x};0)) = (ab)!\left(\frac{1}{b!\, a!^b}-  \frac{1}{a!\, b!^a}\right),
  \end{equation}
  is the difference between the number of $a^b$-partitions and $b^a$-partitions. Some authors have attempted to exploit this fact to prove Foulkes' conjecture (for positive and negative results along these lines see~\cite{wildon,paget,pylyavskyy,sivek}).
  
 It is interesting that we have the following very nice $q$-analog (at $0$) of \pref{dimFoulkes}.
%Evidently, $[n]_q!$ goes to $1$ when we set $q=0$. 
 \begin{proposition} For all $a<b$, we have
\begin{equation}\label{dimqFoulkes}
    \dim(F_{a,b}(\bm{x};q)) = \frac{(ab)!}{1-q}\left(\frac{[b]_q!}{b!}\frac{([a]_q!)^b}{a!^b}-\frac{[a]_q!}{a!}\frac{([b]_q!)^a}{b!^a}\right),
  \end{equation}
and, letting $q\mapsto 1$, we find that
  \begin{equation}\label{dim1Foulkes}
    \dim(F_{a,b}(\bm{x};1)) = \frac{(ab)!(a-1)(b-1)(b-a)}{4}.
  \end{equation}
 \end{proposition}
  \begin{proof}[\bf Proof]
We first calculate $\dim(H_a\circ H_b)$ 
directly as follows

Now, exploiting classical properties of the logarithmic derivative $\mathrm{D}_{\mathrm{log}}\, f:=f'/f$ (with respect to $q$), we easily calculate that 
    $$\mathrm{D}_{\mathrm{log}} \,[b]_q!\,([a]_q!)^b  \Big|_{q=1}= \frac{1}{2}\binom{b-1}{2}+\frac{b}{2}\binom{a-1}{2}.
    $$
From this we may readily obtain that $\lim_{q\to 1}  \dim(F_{a,b}(\bm{x};q)) $ gives~\pref{dim1Foulkes}.
 \end{proof}
  



\newcommand{\sbinom}[2]{\Big(\genfrac{}{}{0pt}{0}{#1}{#2}\Big)}
   %%%%%%%%%%%%%%%%%%%
 \section{The \texorpdfstring{$q$}{q}-conjecture holds at \texorpdfstring{$q=1$}{q}}
We start with an explicit formula that will be helpful in the sequel, setting the simplifying notation
\begin{equation}
    \mathcal{E}_b:=( \left( h_{{2}}+e_{{2}} \right) ^{b}+ \left( h_{{2}}-e_{{2}} \right) ^{b} )/2,\qquad{\rm and}\qquad
    \mathcal{O}_b:=\frac{1}{2}\,\left( ( h_{{2}}+e_{{2}} ) ^{b}- ( h_{{2}}-e_{{2}} ) ^{b} \right),
 \end{equation}
respectively for the even-part and odd-part of $( h_{{2}}+e_{{2}} ) ^{b}$ in the ``variables'' $h_2$ and $e_2$. We thing of these as the homogenous and elementary symmetric functions, for which we have the power-sum expansions $p_1^2=h_2+e_2$ and $p_2=h_2-e_2$.

\begin{lemma}\label{evabk} For all $a$, $b$ and $k$, we have the divided difference evaluation
   \begin{equation}\label{formuledelH}
     \lim_{q\to 1}  \frac{h_1^{abk} -H_a\circ H_b^k}{1-q}= 
    ak\, \sbinom{b}{2}\,h_1^{abk-2}\,e_2+\sbinom{a}{2}\, h_1^{(a-2)bk}\, \mathcal{O}_{bk}.
    \end{equation}
\end{lemma}
\begin{proof}[\bf Proof]
 The limit on the left hand-side is the evaluation at $1$ of the derivative of $H_a\circ H_b^k$ with respect to $q$. We use formula~\pref{defhnpower} to calculate this, exploiting the fact that the evaluation at $1$ of the $q$-derivative of $g(q)=(1-q)^mf(q)$  
is
  $$
       g'(1)=\begin{cases}
      f'(1)& \text{if}\ $m=0$, \\
     - f(1)& \text{if}\ $m=1$, \\
       0& \text{otherwise}.
\end{cases}
  $$
  Using  \pref{defhnpower} and the rules of plethysm to expand $ H_a\circ H_b^k$, 
and observing that the only partitions $\mu$ of $a$ such that $a-\ell(\mu)\leq 1$ are either $\mu=1^a$ or $\mu=21^{a-2}$, we find that we have the expansion
    $$H_a(\bm{x};q)=\frac{[a]!_q}{a!}\, p_1(\bm{x})^a +(1-q) \frac{[a]!_q}{2(a-2)!(1+q)}\,p_1(\bm{x})^{a-2}\,p_2(\bm{x})+ (1-q)^2\,G(\bm{x};q),$$
with $G(\bm{x};1)\not=0$, and where $[a]!_q$ stands for the $q$-analog of $a!$.
Thus, we get that the left hand-side of \pref{formuledelH} evaluates as
{\small \begin{eqnarray*}\label{etape}
 (H_a\circ H_b^k)'(1) &=&\hspace{-0.2cm} \frac{d}{dq}\left(\frac{[a]!_q}{a!}\, p_1^a\circ H_b^k +(1-q) \frac{[a]!_q}{2(a-2)!(1+q)}(p_1^{a-2}\,p_2)\circ H_b^k\right)\Big|_{q=1},\\
  &= &\hspace{-0.2cm}\frac{d}{dq}\left(\frac{[a]!_q}{a!}\,  H_b^{ak}\right)\Big|_{q=1} + \frac{d}{dq}\left((1-q) \frac{[a]!_q}{2(a-2)!(1+q)} \,H_b^{(a-2)k}\cdot (p_2\circ H_b^k)\right)\Big|_{q=1},\\
 &= &\hspace{-0.2cm}\frac{d}{dq}\left(\frac{[a]!_q}{a!}\,  H_b^{ak}\right)\Big|_{q=1} - \left(\frac{[a]!_q}{2(a-2)!(1+q)} H_b^{(a-2)k}\cdot (p_2\circ H_b^k)\right)\Big|_{q=1},\\
&=& \hspace{-0.2cm}\frac{1}{2}\binom{a}{2}p_1^{abk} +ak\,p_1^{(ak-1)b}\frac{d}{dq}(H_b)\Big|_{q=1}- \frac{1}{2}\binom{a}{2}p_1^{(a-2)bk}p_2^{bk},\\
&= &\hspace{-0.2cm}\frac{1}{2}\binom{a}{2}p_1^{abk} +ak\,p_1^{(ak-1)b}\binom{b}{2}\frac{(p_1^{b}-p_1^{(b-2)}p_2)}{2}- \frac{1}{2}\binom{a}{2}p_1^{(a-2)bk}p_2^{bk},
\end{eqnarray*}  }
recalling that $H_n(\bm{x};1)=p_1^n$ and that $p_1^n\circ F= F^n$. Since $p_1=h_1$, $p_1^2=h_2+e_2$ and $p_2=h_2-e_2$, this last expression is clearly equal to the right-hand side of equation~\pref{formuledelH}.
 \end{proof}
It immediately follows that we have the following formula as a difference of two expressions obtained from the lemma.
 \begin{proposition}
   For any $1<a\leq c,d\leq b$ and $k\geq1$, with $n:=abk=cdk$, we have
\begin{equation}
   F_{a,b;c,d}^{(k)}(\bm{x};1)=\left(ak\, \sbinom{b}{2}-ck\, \sbinom{d}{2}\right)\,h_1^{n-2}\,e_2
+\sbinom{a}{2}\, h_1^{(a-2)bk}\, \mathcal{O}_{bk}-\sbinom{c}{2}\, h_1^{(c-2)dk}\, \mathcal{O}_{dk}.
\end{equation}
Moreover, this a positive integer coefficient polynomial in $h_1$, $h_2$ and $e_2$; hence, it expands positively in the Schur basis.
\end{proposition}
\noindent For instance, we get
\begin{eqnarray*}
   F_{2,6;3,4}^{(2)}&=&
 24\,{e_{{2}}}^{12}+252\,{e_{{2}}}^{11}h_{{2}}+1224\,{e_{{2}}}^{10}{h_{
{2}}}^{2}+3868\,{e_{{2}}}^{9}{h_{{2}}}^{3}+7152\,{e_{{2}}}^{8}{h_{{2}}}^{4}\\
&&\qquad +10680\,{e_{{2}}}^{7}{h_{{2}}}^{5}+9744\,{e_{{2}}}^{6}{h_{{2}}}^{
6}+7512\,{e_{{2}}}^{5}{h_{{2}}}^{7}+3192\,{e_{{2}}}^{4}{h_{{2}}}^{8}\\
&&\qquad +1228\,{e_{{2}}}^{3}{h_{{2}}}^{9}+168\,{e_{{2}}}^{2}{h_{{2}}}^{10}+12\,
e_{{2}}{h_{{2}}}^{11}.
\end{eqnarray*}
To make the positivity in the previous proposition more apparent, we exploit the following recursive approach to the calculation of $F_{a,b}=F_{a,b;b,a}^{(1)}$, as a polynomial in $h_1$, $h_2$ and $e_2$, together with Appendix~\ref{AppendixB}.
\begin{proposition}\label{proprec}
 \begin{equation}
   F_{a,b+1}(\bm{x};1)=h_1^a\, F_{a,b}(\bm{x};1)+    
                                         2\,h_1^{(a-2) b}\,\Theta_a(b),
\end{equation}
with $ \Theta_a(b)$ defined recursively as
$$ \Theta_a(b) = (3\,h_2+e_2)\,\Theta_a(b-1)-h_1^2\,(3\,h_2-e_2)\,\Theta_a(b-2)+h_1^4\,(h_2-e_2)\,\Theta_a(b-3),$$
with initial conditions: $\Theta_a(a) = \Theta_{a-1}(a)$, $\Theta_a(a+1) =({a^2e_2}/{2})\, \mathcal{E}_a+({a\,h_2}/{2})\,\mathcal{O}_a$, and
$$ \Theta_a(a+2) = 
\frac{(  \left( a+1 \right) ^{2}-2 )\,e_2}{2} \, \mathcal{E}_{{a+1}}
-\frac{(a+1)\,h_2}{2}\,\mathcal{O}_{{a+1}}+e_2\, ( a\,e_{{2}}+h_{{2}} )\, ( h_2-e_2 )^a.
$$
Moreover, $\Theta_a(b)$ lies in $\N[h_1,h_2,e_2]$.
 \end{proposition} 


\noindent
Using  these calculation techniques, we find that
$$\begin{array}{lll}
F_{{2,3}}=4\,e_2^{3}, &
F_{{2,4}} =8\,e_2^{3}\, ( e_2+2\,h_2 )\\[6pt]& 
F_{{2,5}} =8\,e_2^{3}\,( 2\,e_2^{2}+5\,h_2\,e_2+5\,h_2^{2} ), \\[8pt]
F_{{3,4}}=24\,h_1^{4}\,e_2^{3}\,h_2,&
F_{{3,5}}=8\,h_1^{5}\,e_2^{3}\, ( e_2^{2}+5\,h_2\,e_2+10\,h_2^{2}), \\[6pt]&
F_{{3,6}}=12\,h_1^{6}\,e_2^{3}\, ( e_2^{3}+9\,e_2^{2}\,h_2+15\,e_2\,h_2^{2}+15\,h_2^{3} ), \\[8pt]
F_{{4,5}}=16\,h_1^{10}\,e_2^{3}\, ( e_2^{2}+5\,h_2^{2} ), &
F_{{4,6}}=24\,h_1^{12}\,e_2^{3}\, ( e_2^{3}+4\,e_2^{2}\,h_2+5\,e_2\,h_2^{2}+10\,h_2^{3}), \\[6pt]&
F_{{4,7}}=24\,h_1^{14}\,e_2^{3}\, ( 2\,e_2^{4}+7\,e_2^{3}\,h_2+21\,e_2^{2}\,h_2^{2}+21\,e_2\,h_2^{3}+21\,h_2^{4} ).
\end{array} $$
In particular, for all $b>a>1$, we have.
   $$F_{a,b+1}(\bm{x};1)-h_1^a F_{a,b}(\bm{x};1)\in \N[h_1,h_2,e_2],$$
which implies the analog at $q=1$ of the stability portion of Foulkes' conjecture, namely
\begin{proposition}
    For all $a<b$, and all partition $\lambda$, we have
         $$  \langle\overline{F_{a,b+1}(\bm{x};1)}-\overline{F_{a,b}(\bm{x};1)},s_\lambda(\bm{x})\rangle\in \N.$$
\end{proposition} 
\begin{proof}[\bf Proof]
 Indeed, using the classical Pieri rule for the calculation of $h_1\, s_\lambda$, it is easy to see that
     $$\overline{h_1^a F_{a,b}(\bm{x};1))}-\overline{F_{a,b}(\bm{x};1)}$$
   is Schur positive, since one of the terms in $h_1^a s_\lambda$ is the Schur function indexed by the partition obtained from $\lambda$ by adding $a$ boxes to its first line. Hence the lemma directly implies that
      $$\overline{F_{a,b+1}(\bm{x};1)}-\overline{F_{a,b}(\bm{x};1)}=
          (\overline{F_{a,b+1}(\bm{x};1)}-\overline{h_1^a F_{a,b}(\bm{x};1)})+
          (\overline{h_1^a F_{a,b}(\bm{x};1))}-\overline{F_{a,b}(\bm{x};1)})$$
   is  Schur positive.
\end{proof}
It is interesting to calculate how $F_{a,b}(\bm{x};q)$ expands explicitly as a polynomial in $q$. Indeed, by a direct calculation, one gets
    $$F_{a,b}(\bm{x};q) = (h_b\circ h_a-h_a\circ h_b)+
          ((h_{b-1}\circ h_a)\,h_{a-1}\,h_1 - (h_{a-1}\circ h_b)\,h_{b-1}\,h_1)\,q+\ldots$$
 with similar (but more intricate expressions as illustrated below) for higher degree terms. Hence, the conjectured Schur-positivity of $F_{a,b}(\bm{x};q)$ implies that we have Schur positivity of
      $$ ((h_{b-1}\circ h_a)\cdot h_{a-1} -( h_{a-1}\circ h_b)\cdot h_{b-1})\cdot h_1,$$
but we may show that in fact
\begin{equation}\label{coeffq}
    (h_{b-1}\circ h_a)\cdot h_{a-1} -( h_{a-1}\circ h_b)\cdot h_{b-1}=h_1^\perp (h_b\circ h_a-h_a\circ h_b).
 \end{equation}
 Indeed, it follows readily from the definitions that
    \begin{equation}
       h_1^\perp (h_n \circ h_k) = h_{k-1}\cdot (h_{n-1}\circ h_k).
     \end{equation}
  In a sense this is because $h_1^\perp$ acts as a derivation, sending $h_n$ to $h_{n-1}$, and this is a form of chain-rule.
Hence the positivity of the coefficient of $q$ in $F_{a,b}(\bm{x};q)$ is a consequence of the classical version Foulkes' conjecture, since  Schur positivity is preserved by both operations of multiplication by $h_1$ and its adjoint\footnote{For the usual scalar product on symmetric functions} $h_1^\perp$.
 
However, for higher degree, it does not seem that we can calculate coefficients as easily. To illustrate, we have calculated that the coefficient of $q^2$ in $F_{a,b}(\bm{x};q)$ is equal to
\begin{eqnarray*}
  &&( h_{b-2}\circ h_a)\cdot (h_2\circ h_{a-1}\,h_1)+
        (h_{b-1}\circ h_a)\cdot (h_2\,h_{a-2}+h_1\,h_{a-1}-h_a)-h_b\circ h_a\\
    &&\qquad   - (h_{a-2}\circ h_b)\cdot (h_2\circ h_{b-1}\,h_1)-
        (h_{a-1}\circ h_b)\cdot (h_2\,h_{b-2}+h_1\,h_{b-1}-h_b)+h_a\circ h_b
  \end{eqnarray*}                                  
Property~\pref{coeffq} extends to the wider context of~\pref{doran}, so that the coefficient of $q$ in the right-hand side of~\pref{conjqtDoranformule} (for $k=1$) is indeed Schur positive (assuming that~\pref{doran} holds), since it is equal to
 $h_1\cdot h_1^\perp(h_c\circ h_d-h_a\circ h_b)$.


\section{Expanding Foulkes' conjecture to more general diagrams}
 For partitions $\alpha$, $\beta$, $\gamma$, and $\delta$, none of which equal to $(1)$ and such that $|\alpha|\cdot |\beta|=|\gamma|\cdot |\delta|=n$, let us say that $\foulkes{\alpha}{\beta}{\gamma}{\delta}$ is a \define{Foulkes configuration} for $n$,  if and only if
\begin{equation}\label{Fconf}
   s_\alpha\circ s_\beta\  \preceq_s\  s_\gamma\circ s_\delta.
\end{equation}
Clearly, for $a<b$, Foulkes' conjecture says that $\foulkes{a}{b}{b}{a}$ is a Foulkes configuration. Likewise statement~\pref{doran}, under the conditions there specified, is equivalent to saying that $\foulkes{a}{b}{c}{d}$ is a Foulkes configuration. 
Other cases are possible. Indeed, by direct explicit calculation we find the following:
 $$\begin{array}{lllll}
\foulkes2332,&
\foulkes{11}{111}{3}{11},&
\foulkes{111}{2}{11}{21},&
\foulkes{111}{11}{2}{21},\\[6pt]
\foulkes{2}{4}{4}{2},&
\foulkes{2}{1111}{4}{11},&
\foulkes{11}{4}{31}{2},&
\foulkes{11}{22}{31}{2},\\[6pt]
\foulkes{11}{22}{31}{11},&
\foulkes{11}{31}{211}{2},&
\foulkes{11}{211}{211}{11},&
\foulkes{11}{1111}{31}{11},\\[6pt]
\foulkes{22}{2}{2}{31},&
\foulkes{22}{11}{2}{211},&
\foulkes{211}{2}{11}{31},&
\foulkes{211}{11}{11}{211},\\[6pt]
\foulkes{1111}{2}{2}{31},&
\foulkes{1111}{11}{2}{211},&
\foulkes{2}{5}{5}{2},&
\foulkes{2}{221}{311}{11},\\[6pt]
\foulkes{2}{2111}{311}{11},&
\foulkes{11}{32}{311}{2},&
\foulkes{11}{41}{311}{2},&
\foulkes{11}{11111}{5}{11},\\[6pt]
\foulkes{11111}{2}{2}{311},&
\foulkes{11111}{11}{11}{311}.
\end{array}$$
Under the same assumptions as in~\pref{Fconf} for the partitions involved, we say the we have a \define{$q$-Foulkes configuration} denoted $\qfoulkes{\alpha}{\beta}{\gamma}{\delta}$, if and only if 
 \begin{equation}\label{qfoulkesconfig}
    0\preceq_s \frac{S_\gamma\circ S_\delta-S_\alpha\circ S_\beta}{1-q},
 \end{equation}
with the right-hand side having polynomial coefficients in $q$. 
In particular, this last condition requires that, at $q=1$ we have the equality
   $$ (S_\alpha\circ S_\beta)\Big|_{q=1} = (S_\gamma\circ S_\delta)\Big|_{q=1},$$
 which is equivalent to
\begin{equation}\label{teste}
   e_{\alpha'}\circ e_{\beta'} = e_{\gamma'}\circ e_{\delta'}.
\end{equation}
 For instance, it is easy to check that this last equality holds when
 \begin{equation}\label{bonnecond}
   \alpha=a,\qquad \beta=\underbrace{bb\cdots b}_k,\qquad \gamma=c,\qquad {\rm and}\qquad
        \delta=\underbrace{dd\cdots d}_k,
   \end{equation}
for any $a,b,c,d,k$ in $\N$, such that $ab=cd$, since both sides of~\pref{teste} evaluate to $e_k^{a+b}$. Evidently, all $q$-Foulkes configurations are also Foulkes configurations, but most Foulkes configurations do not satisfy the extra requirement that~\pref{bonnecond} holds. Explicit calculations reveal that this condition significantly reduces the number of possibilities.  

An intriguing development, explicitly checked out for all cases\footnote{Involving $67$ configurations in total.} with $n$ up to $30$, is that having both the necessary conditions~\pref{Fconf} and~\pref{teste} holding seems to be equivalent to having the full $q$-Schur positivity~\pref{qfoulkesconfig} holding too. In other words, we have the following general statement, which would reduce all $q$-versions to the $q=0$ case.

\begin{conjecture}\label{conjstrong}
 For partitions $\alpha$, $\beta$, $\gamma$, and $\delta$, such that $e_{\alpha'}\circ e_{\beta'}=e_{\gamma'}\circ e_{\delta'}$, we have 
\begin{equation}\label{suffcond}
  s_\alpha\circ s_\beta\  \preceq_s\  s_\gamma\circ s_\delta,\qquad \hbox{if and only if} \qquad  0\preceq_s \frac{S_\gamma\circ S_\delta-S_\alpha\circ S_\beta}{1-q}.
\end{equation}
 \end{conjecture}
Clearly, when both $\alpha$ and $\gamma$ are one part partitions, respectively equal to $a$ and $c$, the second condition in~\pref{suffcond} is simply that $e_{\beta'}^a=e_{\delta'}^c$. Observe that this implies that $\beta'=\mu^i$ and $\delta'=\mu^j$ for some partition $\mu$, and $ai=cj$.
Only this simpler version is needed in all cases explicitly calculated. In other words, all configurations that we have found to satisfy $$e_{\alpha'}\circ e_{\beta'}=e_{\gamma'}\circ e_{\delta'}$$ are such that $\alpha$ and $\gamma$ are reduced to one part, and thus of the simple form stated. It seems that this should be easy to prove.
 
 
 

 
    
 

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section*{Appendix A: Background on symmetric functions and plethysm} 
Trying to make this text self-contained, we now rapidly recall most of the necessary background on symmetric functions. For more background, see \cite{bergeron, macdonald}. As is usual, we often write symmetric functions without explicit mention of the variables. Thus, we denote by $p_k$ (as in~\cite{macdonald}) the power-sum symmetric functions
   $$p_k=p_k(x_1,x_2,x_3,\ldots) :=x_1^k+x_2^k+x_3^k+\ldots,$$ 
using which, we can expand the complete homogeneous symmetric functions as 
\begin{equation}\label{hpexpansion}
    h_n=\sum_{\mu\vdash n} \frac{p_\mu}{z_\mu},\qquad {\rm with}\qquad
       p_\mu:=p_1^{d_1}p_2^{d_2}\cdots p_n^{d_n},
   \end{equation}
where $d_k=d_k(\mu)$ is the multiplicity of the part $k$ in the partition $\mu$ of $n$, and $z_\mu$ stands for the integer
      $$z_\mu:=\prod j^{d_j}\,d_j!.$$
 For instance, we have the very classical expansions
     $$h_2=\frac{p_1^2}{2}+\frac{p_2}{2}, \qquad 
         h_3=\frac{p_1^3}{6}+\frac{p_1p_2}{2}+\frac{p_3}{3}.$$
As is also very well known, the homogeneous degree $n$ component $\lambda_n$ of the graded ring $\Lambda$ of symmetric functions, affords as a linear basis the set of Schur functions $\{s_\mu\}_{\mu\vdash n}$, indexed by partitions of $n$. Among the manifold interesting formulas regarding these, we will need the Cauchy-kernel identity.
\begin{eqnarray}
   h_n(\bm{x}\bm{y})&=& \sum_{\mu\vdash n} s_\mu(\bm{x})s_\mu(\bm{y})\label{cauchys}\\
        &=& \sum_{\mu\vdash n} \frac{p_\mu(\bm{x})p_\mu(\bm{y})}{z_\mu},\label{cauchyp}
    \end{eqnarray}
 with $h_n(\bm{x}\bm{y})=h_n(\ldots,x_iy_j,\ldots)$ corresponding to the evaluation of $h_n$ in the ``variables'' $x_iy_j$. Otherwise stated, we
 may express this by the generating function identity
    $$\sum_{n\geq 0} h_n(\bm{x}\bm{y})\,z^n=\prod_{i,j}\frac{1}{1-x_iy_j\,z}.$$ 
\subsection*{Rules of plethysm}\label{plethrules} Plethysm is an associative operation on symmetric functions, characterized by the following properties. Let $f_1$, $f_2$, $g_1$ and $g_2$ be any symmetric functions, and $\alpha$ and $\beta$ be in $\Rat$, then
 \begin{enumerate}\itemsep=6pt
\item[a)] $(\alpha\,f_1+\beta\,f_2)\circ g=\alpha\,(f_1\circ g) +\beta\,(f_2\circ g)$, (left linearity)
\item[b)] $(f_1\cdot f_2)\circ g=(f_1\circ g)\cdot (f_2 \circ g)$,  (left multiplicativity)
\item[c)] $p_k\circ (\alpha\,g_1+\beta\,g_2)=\alpha\,(p_k\circ g_1)+ \beta\,(p_k\circ g_2)$,
\item[d)] $p_k\circ (g_1\cdot g_2)=(p_k\circ g_1)\cdot (p_k\circ g_2)$,
\item[e)] $p_k\circ p_j=p_{kj}$.
  \end{enumerate}
The first four properties reduce any calculation of plethysm to instances of the fifth one. 
For a given symmetric function $f$, one may consider the plethysm $f\circ(-)$ as an operator on symmetric functions. 
In fact, this operator may naturally be extended to any rational fraction in the underlying variables. It is sometimes more convenient to use the alternate notation $f[-]$ for this operator and  ``add'' the further rules 
 \begin{enumerate}\itemsep=6pt
\item[a)] $p_k[g_1/ g_2]=p_k[g_1]/ p_k[g_2]$,
\item[b)] $p_k[x]=x^k$, if $x$ is a variable,
\item[c)] $p_k[c]=c$, if $c$ is a constant.
  \end{enumerate}
Then, considering variable sets as sums $\bm{x}=x_1+x_2+x_3+\ldots$, one observes that $f[\bm{x}]$ corresponds to the evaluation of the symmetric function $f$ in the variables $\bm{x}$. Moreover, Cauchy's formula gives an explicit expression for the expansion of 
             $$h_n[\bm{x}\bm{y}]=h_n[(x_1+x_2+x_3+\ldots)(y_1+y_2+y_3+\ldots)].$$
Likewise $f[1/(1-q)]=f[1+q+q^2+\ldots]$, corresponds to the evaluation $f(1,q,q^2,\ldots)$.

Another interesting classical property of Schur functions may be expressed as
\begin{equation}\label{addition_formula}
   s_\mu[A+B]=\sum_{\nu\subseteq \mu} s_{\mu/\nu}[A]\,s_{\nu}[B],
 \end{equation}
  where $s_{\mu/\nu}$ stand for the \define{skew Schur} function characterized by
      $$\langle s_{\mu/\nu},s_\lambda \rangle= \langle s_\mu,s_\nu^\perp s_\lambda \rangle,$$
writing $\nu\subseteq \mu$ if $\nu_i\leq \mu_i$ for all $i$, and $f^\perp$ standing for the dual operator of multiplication by $f$ for the usual scalar product $\langle-,-\rangle$ on symmetric functions (for which the Schur functions form an orthonormal basis). It is well known that $s_{\mu/\nu}$ is Schur positive, and $s_{\mu/0}=s_\mu$.


\subsection*{Macdonald polynomials} With all this at hand, the polynomial $H_n(\bm{x};q)$ can be explicitly defined as
   \begin{equation}\label{defHn}
       H_n(\bm{x};q):=[n]_q!\, (1-q)^n\,h_n\left[\frac{\bm{x}}{1-q}\right]
   \end{equation}
as before $[n]_q!$  stands for classical the $q$-analog of $n!$:
   $$[n]_q!:=[1]_q\,[2]_q\cdots [n]_q,\qquad {\rm with} \qquad
      [k]_q=1+q+\ldots+q^{k-1}.$$
To get a Schur expansion for $H_n(\bm{x};q)$, we recall the hook length expression
\begin{eqnarray} 
    s_\mu[1/(1-q)] &=&s_\mu(1,q,q^2,q^3,\ldots)\nonumber\\
         &=& q^{n(\mu)}\prod_{1\leq i\leq \mu_j} \frac{1}{1-q^{h_{ij}}},
    \end{eqnarray}
 where  $h_{ij}=h_{ij}(\mu)$ is the hook length of a cell $(i,j)$ of the Ferrers diagram of $\mu$, and $n(\mu):=\sum_{(i,j)} j$. Now, using Cauchy's formula~\pref{cauchys}, with $\bm{y}=1+q+q^2+\ldots$, 
we find that
   \begin{equation}\label{defhnschur}
       H_n(\bm{x};q) =\sum_{\mu\vdash n} \frac{q^{n(\mu)}[n]_q!}{\prod_{1\leq i\leq \mu_j}[h_{ij}]_q}\ s_\mu(\bm{x}).
   \end{equation}
It is well known that the coefficient of $s_\mu(\bm{x})$ occurring here is a positive integer polynomial that $q$-enumerates standard tableaux with respect to the charge statistic. This is the $q$-hook formula. Thus, we find the two expansions.
\begin{eqnarray*}
     H_3(\bm{x};q)&=& \frac{[2]_q[3]_q}{6}\,p_1(\bm{x})^3+\frac{[3]_q}{2}(1-q)\,p_1(\bm{x})p_2(\bm{x})+\frac{[2]_q}{3}(1-q)^2\,p_3(\bm{x}),\\[4pt]
           &=& s_3(\bm{x})+(q+q^2)s_{21}(\bm{x})+q^3s_{111}(\bm{x}).
             \end{eqnarray*}
It is clear that $H_n(\bm{x};0)=h_n$. The $H_n(\bm{x};q)$ function encodes, as a Frobenius transform, the character of several interesting isomorphic graded $\SymGroup_n$-modules such as:  the coinvariant space of $\SymGroup_n$, the space of $\SymGroup_n$-harmonic polynomials, and the cohomology ring of the full-flag variety. More precisely, this makes explicit the graded decomposition into irreducibles of these spaces. Thus, the coefficient of $s_\mu(\bm{x})$ in formula~\pref{defhnschur} corresponds to the Hilbert series\footnote{Graded dimension.} of the isotropic component of type $\mu$ of this space. Using \pref{cauchyp} to expand $H_n$, the global Hilbert series of these modules can be simply obtained by computing the scalar product
\begin{eqnarray}\label{hilbcoinv}
   \langle p_1^n,H_n\rangle&=&\sum_{\mu\vdash n} \langle p_1^n,s_\mu\rangle\, q^{n(\mu)}   [n]_q!\prod_{(i,j)\in \mu} \frac{1-q}{1-q^{h_{ij}}},\\
     &=&\sum_{\mu\vdash n}  \langle p_1^n,p_\mu\rangle \frac{p_\mu(1/(1-q))}{z_\mu}\prod_{k=1}^n (1-q^k)\\
     &=& \left(\frac{1}{1-q}\right)^n \prod_{k=1}^n (1-q^k)\\
     &=& [n]_q!.
      \end{eqnarray} 
To see this, recall that  $\langle p_\mu,p_\lambda\rangle$ is zero if $\mu\not=\lambda$, and $\langle p_\mu,p_\mu\rangle=z_\mu$. To complete the picture, let us also recall that $\langle p_\mu,s_\lambda\rangle$ is equal to the value, on the conjugacy class $\mu$, of the character of the irreducible representation associated to $\lambda$. In particular, it follows that
\begin{equation}
     H_n(\bm{x};1)=h_1^n= \sum_{\mu\vdash n} f_\mu\, s_\mu(\bm{x}).
  \end{equation}
This is the Frobenius characteristic of the regular representation of $\SymGroup_n$, for which the multiplicities $f_\mu$
are given by the number of standard Young tableaux of shape $\mu$.
Beside this notion of Frobenius transform that ``formally'' encodes $\SymGroup_n$-irreducibles as Schur function, another more direct interpretation of the above formulas is in terms of characters of polynomial representations of $\GL(V)$, with $V$ an $N$-dimensional space over $\mathbb{C}$. Recall that the character, of a representation $\rho:\GL(V)\rightarrow \GL(W)$, is a symmetric function $\chi_\rho(x_1,x_2,\ldots,x_N)$ of the eigenvalues of operators in $\GL(V)$.  Through Schur-Weyl duality, out of any $\SymGroup_n$-module $R$ and any $\GL(V)$-module $U$, one may construct a representation of $\GL(V)$:
          $$R(U):=R\otimes_{\C\SymGroup_n} U^{\otimes n},$$
 where $\SymGroup_n$ acts on $U^{\otimes n}$ by permutation of components. This construction is functorial:
      $$R:\hbox{$GL(V)$-Mod}\longrightarrow  \hbox{$GL(V)$-Mod},$$
and the character of $R(U)$ is the plethysm $(f\circ g)(x_1x_2,\ldots,x_N)$, where $f$ is the Frobenius characteristic of $R$ and $g$ the character of $U$.  Furthermore, under this construction, irreducible polynomial representations of $\GL(V)$ correspond to irreducible $\SymGroup_n$-modules $R$. If such is the case, one writes $S^\lambda(V)$ when $R$ is irreducible of type $\lambda$. The corresponding character is the Schur function $s_\lambda(x_1,x_2,\ldots,x_N)$. For the special case $\lambda=(n)$, we get the symmetric power $S^a(V)$ whose character is $h_a(x_1,x_2,\ldots,x_N)$, hence the character of $S^a(S^b(V))$ is the plethysm $h_a\circ h_b$. 


%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section*{Appendix B: $\N$-positivity of $\Theta_a(b)$}\label{AppendixB}
The proof of $\N$-positivity of the solution of the recurrence occurring in Proposition~\ref{proprec} may be directly obtained as follows. Let us consider the positive integer coefficient series\footnote{Which is obviously terminating when $a$ is an integer.} defined as:
 \begin{equation}
     \rho(z;a):=\sum_{k=1}^\infty k\,a\binom{a+1}{2k+1}\, z^{2\,k+1},
 \end{equation}
and then define recursively
$\theta_n(z)=\theta_n(z;a)$ as follows
 $$\theta_n(z)=\left( 3+z \right) \theta_{n-1}(z)+ \left( 1+z \right)  \left( z-3
 \right) \theta_{n-2}(z)+ \left( 1+z \right) ^{2} \left( 1-z
 \right) \theta_{n-3}(z),$$
 with initial conditions $\theta_0(z):=\rho(z;a-1)$, $\theta_1(z):=\rho(z;a)$, and
  $$\theta_2(z) = \sum_{k=1}^\infty\left( k(a-1)\binom{a+2}{2k+1}\, z^{2k+1} +
2k \binom{a+1}{2k+1}\, (1+z)\,z^{2\,k+1}\right).$$
For any $a>2$ (in $\N$), $\theta_{n}(z;a)$ is clearly a degree $a+n$ polynomials in the variable $z$, with positive integer coefficients. This makes it obvious that the expression $\Theta_a(b)$ occuring in Proposition~\ref{proprec} is $\N$-positive, since
     $$\Theta_a(b) =h_2^b\, \theta_{b-a}(e_2/h_2;a).$$
   
 
\subsection*{Acknowledgements}
Thanks to Mark Wildon for making publicly available his extensive table of calculations on Foulkes' conjecture:

\begin{center}
\href{http://www.ma.rhul.ac.uk/~uvah099/other.html#FoulkesData}{http://www.ma.rhul.ac.uk/$\sim$uvah099/other.html\#FoulkesData}
\end{center}
These were very helpful for our calculations which also made use of John Stembridge's SF package. Similar tools are available in Sage. Thanks to the anonymous referee for his/her suggestions, and for drawing our attention to reference~\cite{Kahle}. Using the results therein, it seems that much larger experiments could be calculated for some of our conjectures.
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\end{document}


 

	
	

 